Author's response to reviews

Title: Aspirin for the prevention of cognitive decline in the elderly: rationale and design of a neuro-vascular imaging study (ENVIS-ion)

Authors:

Christopher M Reid (chris.reid@monash.edu)
Elsdon Storey (elsdon.storey@monash.edu)
Tien Y Wong (ophwty@nus.edu.sg)
Robyn Woods (robyn.woods@monash.edu)
Andrew Tonkin (andrew.tonkin@monash.edu)
Jie Jin Wang (jiejin.wang@monash.edu.au)
Anthony Kam (a.kam@alfred.org.au)
Andrew Janke (andrew.janke@cai.uq.edu.au)
Rohan Essex (rohanessex@hotmail.com)
Walter P Abhayaratna (walter.abhayaratna@act.gov.au)
Marc M Budge (marc.budge@anu.edu.au)
the ASPREE Study Group (not@valid.com)

Version: 4 Date: 1 February 2012

Author's response to reviews:

In Manuscript "Background" changed ischemia to ischaemia x 3.
In "Methods/design" changed anemia to anaemia.